Author's response to reviews

Title: Core competencies in the science and practice of knowledge translation: description of a Canadian strategic training initiative

Authors:

sharon e straus (sharon.straus@utoronto.ca)
melissa brouwers (mbrouwer@mcmaster.ca)
david johnson (david.johnson@albertahealthservices.ca)
john lavis (lavisj@mcmaster.ca)
france legare (france.legare@fmed.ulaval.ca)
sumit r majumdar (me2.majumdar@ualberta.ca)
k. ann mckibbon (mckib@mcmaster.ca)
anne sales (anne.sales1@gmail.com)
dawn stacey (dawn.stacey@uottawa.ca)
gail klein (kleing@smh.ca)
jeremy grimshaw (jgrimshaw@ohri.ca)

Version: 5 Date: 23 November 2011

Author's response to reviews:

Thank you for providing the edited version of the manuscript. All of the changes have been accepted and a clean copy of the manuscript has been prepared.